

Opportunities to improve palliative care: towards a more patient-centred and proactive approach

Verhoef, M.J.

Citation

Verhoef, M. J. (2022, November 23). *Opportunities to improve palliative care: towards a more patient-centred and proactive approach*. Retrieved from https://hdl.handle.net/1887/3486999

Version:	Publisher's Version
License:	<u>Licence agreement concerning inclusion of doctoral</u> <u>thesis in the Institutional Repository of the University</u> <u>of Leiden</u>
Downloaded from:	https://hdl.handle.net/1887/3486999

Note: To cite this publication please use the final published version (if applicable).



CHAPTER 8

General discussion

This chapter consists of five parts. Part 1 provides the main findings of the studies presented in this thesis. Part 2 reflects on the methodologies used in our studies. Part 3 describes the implications of the outcomes of this thesis regarding a proactive approach in palliative care. Part 4 includes recommendations for future research and Part 5 provides recommendations for education, clinical practice, and policy.

8.1 MAIN FINDINGS

Objective 1: To assess the extent to which end-of-life care is taught at medical schools in the Netherlands and to find opportunities to improve Dutch medical curricula.

Chapter 2 describes a cross-sectional study assessing the extent to which end-of-life care was part of Dutch medical curricula. This study was focused on end-of-life care as essential part of palliative care. The study was conducted during the academic year of 2015-2016. A checklist including the essential domains of end-of-life care education was constructed based on literature. The study demonstrated that the national blueprint on medical education included four of the five domains of end-of-life care. One faculty taught an elective course that included all essential domains. None of the eight medical faculties taught all domains of end-of-life care; these domains were taught in the light of other courses but did not specifically address end-of-life care.

Objective 2: To explore palliative care needs and the extent of proactive care in patients with advanced cancer who visited the emergency department (ED) in the last three months of their lives.

Knowledge about problems leading to visits to the Emergency Department (ED) in patients in the last three months of their lives and about who are at high risk of approaching death is relevant in ED-triggered palliative care. In **Chapter 3**, we conducted a mortality followback study in 420 patients with advanced cancer who visited the ED up to three months before they died. Our study showed that their care was often still focused on diseasemodification. Only a few patients had limitations on life-sustaining treatments. This may have led to a high percentage of hospitalisations and in-hospital deaths in this study. Factors associated with approaching death were lung cancer, neurologic deterioration, dyspnoea, hypercalcemia, and jaundice.

Objective 3: To describe the end-of-life trajectory and quality of care of patients with a haematological malignancy who visited the ED in the last three months of their lives, compared to patients with advanced cancer.

Patients with a haematological malignancy often receive more aggressive end-of-life care than patients with a solid tumour. Insight into cues for proactive care can help improving ED-triggered palliative care in patients with a haematological malignancy. **Chapter 4** presents a mortality follow-back study in 78 patients with a haematological malignancy and 420 patients with a solid tumour visiting the ED in the last three months of life. Before their ED-visits, patients with a haematological malignancy had less often discussed limitations on life-sustaining treatments. Since their ED-visit, patients with a haematological malignancy were more often hospitalised after their ED-visit, received more aggressive end-of-life care, and more often died in-hospital, in the intensive care unit or in the ED, compared to patients with a solid tumour.

Objective 4: To evaluate the performance of the surprise question to identify palliative care needs in patients with advanced cancer visiting the ED.

The Surprise Question (SQ), "Would I be surprised if this patient died in the next 12 months?", is an instrument to identify patients with palliative care needs. A meta-analysis of Downar et al. demonstrated that the SQ lacks sensitivity and may be more accurate when combined with other indicators of palliative care needs. **Chapter 5** describes an observational cohort study in 245 patients with advanced cancer visiting the ED in 2013 and 2014. The SQ had the following test characteristics: sensitivity of 89%, specificity of 40%, positive predictive value of 85%, negative predictive value of 50% and a c-index of 0.56. In patients in whom physicians would not be surprised if they died within one year, ECOG performance status 3-4 was an independent predictor for approaching death. Addition of ECOG performance status 3-4 as a second step to the SQ improved the c-index (0.65), specificity (92%) and positive predictive value (95%) at cost of sensitivity (40%) and negative predictive value (29%).

Objective 5: To explore the association between symptom burden and information needs of patients referred to a hospital palliative care consultation team using the Leiden Guide on Palliative care (LGP).

The LGP is a conversation guide on palliative care consisting of two parts. The first part comprises the Utrecht Symptom Diary, a translated and adapted Edmonton Symptom Assessment Scale.¹ The second part is a question prompt list on palliative care, a Dutch adaption of the version developed by Clayton et al.² The relationship between patient-reported symptom burden and information needs has not been studied before. **Chapter**

6 includes an observational study in 321 patients referred to a palliative care consultation team who had used a LGP between 2013 and 2018. We found that patients reported highest median symptom burden for *Fatigue* (7; range 4-8) and *Less appetite* (6; range 3-9). Most information needs were about *Fatigue* (68.0%), *Possibilities for managing complaints in the future* (68.0%) and *Complaints expected for the future* (67.3%). Patients had significantly more information needs about symptoms for which they reported clinically relevant burden, or which they had prioritised. However, patients who reported mild symptom burden, also regularly indicated that they wanted information about how they can manage care at home or in a hospice, while patients in the early palliative phase wanted more information about how treatment can affect their quality of life.

Objective 6: To evaluate and further develop the question prompt list of the LGP to prepare the question prompt list for use by generalist palliative care clinicians.

In **Chapter 7**, we performed a mixed-methods study in 2017-2018 to evaluate use of the question prompt list of the LGP and to further develop it using 35 interviews with patients, family, and generalist clinicians and 32 audiotaped consultations. Patients, family, and clinicians regarded the question prompt list as relevant. Patients and family explained that using the question prompt list before the consultation structured their thoughts and helped them ask their questions during consultations. It supported them to regain a sense of control. Although using the question prompt list could evoke strong emotions, they indicated that their real challenge was to accept being a patient in the palliative phase. Clinicians found that the question prompt list could support them as a reminder of discussion topics. During consultations in the hospital with palliative care consultants, topics patients and family had indicated were discussed frequently, but also topics aimed at (re)organising life at home were discussed.

8.2 REFLECTIONS ON THE USED METHODOLOGIES

In **Chapter 2**, a cross-sectional study was conducted to acquire an overview of the status of education of end-of-life care in Dutch undergraduate medical curricula. A checklist including essential domains of end-of-life care education was established based on literature review to assess the national blueprint on medical education and to form structured questionnaire to send to study coordinators. Using these methods, results of this study represent the state of education on end-of-life care in Dutch medical education in the academic year of 2015-2016 making use of internationally endorsed domains. Some participants indicated it was difficult to point out which curricular parts

Chapter 8

were actually about end-of-life care. They felt that end-of-life care, if taught, seemed to be intertwined with other topics. This may have biased the results in both positive and negative directions. To minimise this effect, all respondents were interviewed about their answers in the questionnaire. Data were provided of 15 out of 16 bachelor and master curricula in the Netherlands. Unfortunately, data from one master curriculum were not provided despite repeated reminders. Since data from the 15 other curricula were collected and analysed, this study still provides a reliable overview of end-of-life care in Dutch undergraduate medical curricula in 2015-2016. All elective courses were assessed by the researchers using the checklist. Any electives that were not found in study brochures were therefore not included, but it seems unlikely any electives were missed.

In **Chapter 3** and **Chapter 4**, we performed a mortality follow-back study in patients with advanced cancer or a haematological malignancy visiting the Emergency Department (ED) in the last three months of their lives in 2011-2013. A follow-back timeframe of three months was chosen because a life-expectancy of three months or less is an indication for referral to palliative-terminal care in the Netherlands. Including patients who died up to three months after their ED-visit gives pragmatic insight into the end-of-life trajectories of these patients, and into the intensity of care provided to them. Because we collected data retrospectively, there is a risk of registration bias and unmeasured confounding. Our study design choice for a mortality follow-up back study instead of a retrospective cohort study including all visiting the ED in 2011-2013 has important consequences. Since those who did not die within 3 months were not included, analysis of risk factors for approaching death after the ED-visit are only representative for those who will eventually die within three months.

In **Chapter 5**, the performance of the Surprise Question (SQ) in patients with advanced cancer visiting the ED is evaluated using a prospective observational design. E-questionnaires including the SQ were sent to attending physicians within one working day from the patient's ED-visit. Although e-questionnaires were sent as soon as possible, there might be some recall bias from attending physicians particularly remembering patients who were sicker than other patients. Patient data were retrospectively collected from their charts, which may have introduced bias by under-registration of characteristics and symptoms. Under-registration of symptoms that are prevalent in the palliative phase is possible because these symptoms are not routinely screened in our ED.

Chapter 6 describes an observational study on symptom burden and information needs of patients referred to a hospital palliative care consultation team between 2013 and 2018 and who had used the Leiden Guide on Palliative care (LGP). In the LGP, the Utrecht

Symptom Diary precedes the question prompt list. Because of this order, patients may have indicated information needs about symptoms more frequently than information needs about other topics. This study included mostly patients with advanced cancer, reflecting a specialised setting of an academic medical centre. Therefore, caution must be taken when interpreting the results for patients with other diseases in the palliative phase. The LGP is selectively distributed by the palliative care consultants of Leiden University Medical Center, who estimate whether patients are fit enough to go through the LGP. If they already may be in the dying phase, the LGP is not handed out. This may have led to selection bias because the study lacks data from those in the dying phase and those who did not have enough energy to go through the LGP. In addition, it is possible that some patients did not receive or use the Leiden Guide because they were not emotionally ready to read about palliative care topics. Patients could decide themselves not to use (parts of) the LGP.

In **Chapter 7**, a mixed-methods research design was used to evaluate how the question prompt list of the LGP is used and could be further developed. Data from interviews with patients, family and non-specialist palliative care clinicians, and audiotaped consultations of our palliative care consultation team were included. The patients who were interviewed mostly had cancer. It is possible that patients with different diagnoses have other information needs. The included generalist (non-specialist) clinicians had not used the question prompt list before. Their responses in the interviews were therefore hypothetical and may differ from clinical practice.

8.3 IMPLICATIONS OF THE OUTCOMES OF THIS THESIS

The studies presented in this thesis address various themes within the broad scope of palliative care. The common theme explored is proactivity in palliative care, and how it can be improved. Opportunities are identified to improve a proactive approach in palliative care in good quality undergraduate medical education, empowerment of patients and family during palliative care consultations, timely identification of palliative care needs and in the use of a two-track approach for patients with unpredictable yet life-threatening diseases. This section includes four paragraphs:

- 1. Definition and meaning of proactive palliative care
- 2. Promoting knowledge and self-initiation
- 3. Intention to produce good results and avoid (future) problems
- 4. Thinking ahead to be able to act before things happen

0

The first paragraph explains more about what proactivity and proactive palliative care entail. The following three paragraphs elaborate on the characteristics of proactive palliative care that are derived from paragraph 1 and studied in this thesis.

8.3.1 Definition and meaning of proactive palliative care

Proactivity is an integral part of palliative care. When searching for the definition of proactivity in English dictionaries, the following definitions of '*proactivity*' can be found:

- Taking action by causing change and not only reacting to change when it happens (Cambridge dictionary)
- Intending or intended to produce a good result or avoid a problem, rather than waiting until there is a problem (American dictionary)
- Taking action to make changes yourself rather than reacting to things that happen (Business English dictionary)

From these definitions of proactivity, the following characteristics of a proactive person ('the actor') can be derived:

• the actor must have **knowledge and self-initiation** to act himself and **take control**.

• the actor must have an intention to make a change to have good results or to avoid problems;

the actor must think ahead to be able to act before things happen;

The concept of proactivity is studied in-depth in management and business studies and can be regarded as a process in which goals are proactively set and strived for.³ In health care, the concept of proactivity is not widely studied yet. Proactivity is mainly described as patient behaviour, a characteristic or behaviour of (successful) healthcare professionals and as health promotion.⁴⁻⁸ In these studies, proactivity was defined as: anticipating in correcting problems before they "become insurmountable mountains"; information-seeking, advice-seeking, participating in decision-making, assertiveness towards healthcare professionals, actively taking care of one's health status, promoting health and wellbeing. 'The actor' in health care can refer to patients, their family, and clinicians.

Palliative care is deemed appropriate already early in the illness trajectory concurrent with life-prolonging treatments. Palliative care models such as illustrated by Murray et al. show that patients and family have care needs in physical, psychological, social, and spiritual domains during the whole illness trajectory (Fig.1 in Chapter 1).^{9, 10} Palliative care integrated early into standard life-prolonging or even curative therapy creates the opportunity to support patients and family to express what they value in life and how they want to live their lives. In this way, appropriate care can be organised

concordant with the patient and family's needs and wishes. They can timely prepare for the end of life. In scientific literature, proactive palliative care is often equated to early palliative care, and involves timely identification of patients with palliative care needs, proactive conversations about patients' and family's wishes about the end of life, patient empowerment and prevention of future problems.^{11, 12} However, proactive palliative care not only refers to early palliative care, but is an appropriate approach along the whole illness trajectory. Sometimes, dependent on the illness trajectory, it is not possible to initiate palliative care early; in these cases, a 'late' proactive approach is also demanded to achieve person-centred and appropriate care.

Proactive palliative care includes actions such as goals-of-care discussions, multidimensional treatment of symptoms and supportive care of patient and family, multidisciplinary collaboration, and preventive medicine. Other aspects of proactive palliative care are communication with other healthcare providers about end-of-life issues and making and sharing proactive care plans.¹³⁻¹⁸ Palliative care can therefore be considered as an active approach to care in the last phase of life with proactive characteristics. The 2017 Netherlands Quality Framework for Palliative Care indeed describes that one of the core values of palliative care is being proactive: *"Together with the patient and family, the physical, psychological, social and spiritual dimensions should be* **proactively** assessed and documented for situations in which:

- normal, everyday attention is enough;
- there is a need for guidance or treatment;
- crisis intervention is necessary."^{19, 20}

In proactive palliative care, goals of care should be discussed in shared decision-making. Aligning with the 2002 World Health Organisation's definition of palliative care and the previously derived definition of proactivity, proactive palliative care has the following goals:^{3,21}

to produce a good result, i.e., to improve or maintain quality of life and appropriate care.²² Appropriate care is associated with good supportive care, care decisions including waiving or stopping treatment, sufficient symptom relief, care at home, following wishes of the patient, and effective communication (including advance care planning, right attitude, listening and informing).²³ To achieve this, palliative care should be of good quality and in line with preferences of patients and family, while taking into account the four dimensions of palliative care (psychological, social, physical and spiritual), and stimulate self-management of patients and family. To optimally benefit from palliative care, a palliative care approach should be integrated into standard care early in the disease trajectory. Along the whole illness trajectory,

good quality palliative care is proactive rather than reactive. A proactive attitude is preferred to prevent problems and suffering in the (near) future.

 to avoid (future) problems, i.e., to avoid poor quality of the last phase of life. A poor quality of the last phase of life includes unnecessary symptom burden, caregiver burden and inappropriate care. Inappropriate care is associated with insufficient supportive care, overtreatment and insufficient symptom relief, hospital care, not following wishes of the patient, and poor communication (mainly informing insufficiently and not-listening).²³

Taking the aforementioned concepts and definitions of proactivity and palliative care into account, the following themes can be identified, with the following recommendations articulated in this thesis:

Theme	Subthemes addressed in this thesis
8.3.2 Promoting knowledge and self- initiation in clinicians, patients, and family	 Good education for undergraduate medical students Patient and family empowerment
8.3.3 Intention to produce good results and avoid (future) problems	 Quality indicators for palliative care in patients with a haematological malignancy Timely identification of palliative care needs: ED-triggered palliative care in patients with advanced cancer
8.3.4. Thinking ahead to be able to act before things happen	1. A two-track approach 2. Discussing future scenarios

The next sections will discuss how the studies included in this thesis contribute to the concept of proactive palliative care, and which recommendations can be made based on this thesis.

8.3.2 Promoting knowledge and self-initiation in clinicians, and patients and family

1. Good education for undergraduate medical students

(partly adapted from de Bruin et al. NED TIJDSCHR GENEESKD. 2020;164:D4115)²⁴

In the Netherlands, provision of palliative care is organised following the generalist – specialist palliative care system.²⁵ This means that all clinicians should have at least basic palliative care competences. These responsibilities are described in the Netherlands Quality Framework for Palliative Care 2017.¹⁹ Furthermore, a publication in JAMA about the need for appropriate palliative care during the COVID-19 pandemic, reminded that palliative care is a human right, and that patients without curative options should not be abandoned.²⁶ The study in **Chapter 2** addressed undergraduate medical education.

General discussion

The study demonstrates that the 2009 Dutch national blueprint on medical education insufficiently mentioned end-of-life care, and none of the formal undergraduate curricula of the Dutch medical schools in the academic year of 2015-2016 offered all essential elements of end-of-life care education. The national blueprint on medical education is authoritative for the curricula for Dutch medical faculties and should therefore at least include the essential domains of end-of-life care, which were found using a literature review. Pieters et al. found that final year medical students indicated that various topics were insufficiently addressed in their curricula and 60% of them felt unconfident in providing palliative care.²⁷ In March 2020, the new blueprint on medical education was published.²⁸ The introduction of the blueprint stated that it aims to prepare medical students for the future, including challenges of ageing populations and multimorbidity. The definition of palliative care is added, and generalist palliative care is now anchored as basic knowledge any freshly graduated physician should have.

Chapter 2 provides insight in how individual bachelor and master curricula in the Netherlands can improve their programmes so that young physicians are sufficiently prepared for clinical practice. Many efforts have been made to integrate palliative care into existing Dutch medical curricula. The Palliative care Alliance Sharing Educational tools for Medical student Competencies development (PASEMECO) project, listed six Entrustable Professional Activities (EPAs) for undergraduate medical education regarding palliative care that should be taught to all medical students.²⁹ These EPAs reflect which clinical activities young physicians can encounter in clinical practice. Additionally, the PASEMECO project offers an online toolbox with elements of palliative care education which are published on the Palliaweb website.³⁰ These elements are collected from all Dutch universities and are free to use by teachers.

This thesis only addressed undergraduate medical education. The O2PZ programme, which is supported by the Dutch government since 2019, aims at securing palliative care education at all educational levels in the Netherlands. In this programme, palliative care specialists, teachers and educationalists cooperate in developing education frameworks that provide an overview of all continuing education courses, designing and implementing education, and increasing the visibility of what is happening to optimise palliative care education.³¹ In the O2PZ programme, the EPAs for undergraduate medical education formulated by the PASEMECO project are further developed and complemented.³² Amongst many other activities, the O2PZ programme supports medical faculties in implementing palliative care education, so that their curricula meet the criteria of the blueprint.

Integration of palliative care and end-of-life care education into undergraduate medical curricula contributes to the preparation of future medical doctors to provide generalist palliative care. Promoting their knowledge and skills is a prerequisite for employing a proactive palliative care approach in clinical practice.

2. Patient and family empowerment

This paragraph demonstrates how this thesis articulates to the concept of patient empowerment as part of the person-centred care model. Proactive palliative care includes the improvement of patients and family's self-initiation: empowerment. Part of empowerment is the exploration of the values and needs of individual patients, which is also essential in person-centred care. Many definitions of person-centred or patient-centred care exist and are often used interchangeably. Little et al. identified core activities of patient-centred care:^{30, 33}

- Exploring how patients experience their disease and illness, including their ideas and feelings about it, their expectations of the consultation and how their functioning is affected;
- Understanding the patient as a whole, including personal and developmental aspects and their context;
- Establishing a partnership with the patients, sharing problems, priorities, and goals
 of treatment, agreeing on the patient and clinician's roles, enhancing this patientclinician relationship by sharing power and sustaining a relationship that is caring
 and healing;
- Promoting health, including enhancement of health, reduction of risks and the early detection of disease.

Person-centred care originates from care for people with a chronic illness, where planned care (i.e., proactive care) is preferred to reactive care because it creates more opportunities to provide appropriate care. A person-centred approach is then appropriate since the person himself is the expert of his own experience of being ill. It is patient-friendly to support them so that they can manage their needs in a way that fits their way of life and being best.³⁴ Person-centred care has favourable patient outcomes: patients have a better understanding of their illness, are more aware of their health and treatment options, feel more capable and confident to make decisions, and know more about symptoms.³⁴⁻³⁶

Person-centred care is a model of care that makes it possible to share knowledge, status, and decision-making in an equitable patient-clinician relationship. McWilliam et al. argue that these are elements that support the empowerment of patients and family.³⁷ The 'power' in a patient-clinician relationship changes when the 'expert' role of being ill is balanced between the clinician and the patient by sharing knowledge and experiences,

and providing support. This is in line with the Netherlands Quality Framework on palliative care, which states that shared decision-making in palliative care is *"the continual process in which care is tailored to the personal situation and achievable values, wishes and needs of the patient and family"*.¹⁹ A power shift in the patient-clinician relationship towards patients and family is therefore desirable and warranted.

Patient empowerment is defined in various ways.³⁸ One of the definitions is that patient empowerment is "an approach to health promotion involving patient autonomy, ensuring the patient is actively involved in their care, relying on an achievement of self-efficacy or having a sense of control in one's life."³⁴ Empowering a patient can mean that the patient had been disempowered in the first place. Disempowerment may hinder their participation in decision making about the palliative phase. This experience of disempowerment corresponds with Rolland's 'crisis phase' of the 'Time Phases of Illness' in his integrative treatment model.³⁹ In this model, patients and family go through steps aimed at socialization to the new palliative diagnosis, which includes seven developmental tasks patient and family need to go through. Examples of these tasks are developing a meaning of the illness so that patients and family can stay in control; accepting that the illness is permanent; and learning to live with the life-limiting illness and problems associated with the illness. Patients and family are vulnerable during these developmental tasks and need advice from the clinicians to complete them. Effective communication which effectively enhances empowerment has several benefits for patients.⁴⁰ It is demonstrated that patients found they were better able to adapt to losses, including preparing for deterioration, management of personal issues and changing of priorities in life. They felt they were better capable of continuing their lives.⁴¹

The study in **Chapter 7** demonstrates that patients lost empowerment at the moment they found out they were in the palliative phase of their disease and that they felt more in control by using the question prompt list. They knew which issues were on their minds and which questions they wanted to ask. They felt they were better able to manage their lives because they could ask purposeful questions. As a result, the clinician's role shifts from discussing what they think are preferences of patients and family, to helping patients and family discovering their own preferences for care. **Chapter 6** shows that a question prompt list is a valuable addition to assessing symptoms, because it allows patients and family to not only indicate their current problems, but also their worries. These results show that our question prompt list enhances patient and family empowerment.

Non-specialist clinicians indicated in **Chapter 7** that using a question prompt list provides them with an overview of the information needs of patients and family and may help deepening their consultations. In this way, clinicians can give them tailored advise, so 8

that patients and family can decide for themselves. This corresponds with Little et al.'s definition of patient-centred care: exploring patient's questions, understanding the patient within their context, enhancing the patient-clinician relationship and promoting health in line with the wishes of patients and family.³³ Furthermore, non-specialist clinicians indicated that the question prompt list can be helpful in their practice.

Because patient and family empowerment is desirable and warranted, the use of a question prompt list in palliative care is recommended. A question prompt list supports a proactive palliative care approach from a person-centred perspective, given that patient-related and clinician-related barriers are taken into account (Chapter 7, Table 3). This thesis demonstrates that the combination of the assessment of symptom burden and the assessment of information needs provides more opportunities for a person-centred and proactive approach than assessment of symptom burden or information needs alone. Use of the Leiden Guide on Palliative Care, which includes both, is therefore recommended in discussing and organising proactive palliative care.

8.3.3. Intention to produce good results and avoid (future) problems

1. Quality indicators for palliative care in patients with a haematological malignancy

Palliative care, especially early palliative care, leads to favourable outcomes compared to standard care in patients with advanced diseases, such as improved quality of life and relief of symptom burden.⁴²⁻⁴⁹ These outcomes should be the goal of investing in producing good results and avoiding (future) problems. An example of a 'problem' that should be avoided is hospitalisation in the last months of life, because this can lead to aggressive or potentially inappropriate end-of-life care.⁵⁰ Research has shown that not all end-of-life hospitalisations are necessary.⁵¹ Hospitalisations in the last three months of life can be avoided in 24%, according to general practitioners.⁵²

Indicators for aggressive end-of-life care are described in more detail in **Chapter 1**. Indicators in the study on patients with cancer described in **Chapter 4**, by Earle et al., are the following:

- Receiving chemotherapy in the last 14 days of life;
- Starting a new chemotherapy regimen in the last 30 days of life;
- >1 emergency room visit in the last month of life;
- >1 hospitalisation in the last month of life;
- Admission to the intensive care unit in the last month of life;
- Death in an acute care hospital;
- Lack of admission to hospice;
- Admission to hospice <3 days before death.^{53, 54}

In **Chapter 4**, the causes of death of patients with a haematological malignancy were disease progression (46%), treatment toxicity (40%), or both (9%), illustrating this particular endof-life trajectory. These illness trajectories are often unpredictable with a sudden decline while patients are undergoing intensive treatment with curative intent. Because of the typical illness trajectory, one may wonder whether the common indicators of Earle et al. are appropriate for measuring the quality of end-of-life care in patients with a haematological malignancy.⁵⁵ Haematologists do not necessarily believe hospice admission early in the illness trajectory is an indicator of good quality of end-of-life care, because hospices or hospice care often cannot provide blood transfusions in cases of severe symptoms of anaemia, such as exhaustion, dyspnoea and bleeding.⁵⁶ More often, patients with a haematological malignancy will choose the possibility of blood transfusions over hospice care, risking of hospitalisations and admission to the ICU.⁵⁶ A study by Odejide et al. showed that haematologists found that the following quality indicators were considered acceptable by haematologists:

- Hospice admission > 7 days before death;
- No chemotherapy ≤ 14 days before death;
- · No intubation in the last 30 days of life;
- No cardiopulmonary resuscitation in the last 30 days of life.⁵⁷

These indicators for quality of care represent events that should be prevented, but do not explain what quality palliative care looks like. A two-track approach can support a proactive palliative care approach in haematology. Figure 1 in **Chapter 4** illustrates that curative care and palliative care should be concurrent because patients can have needs from both tracks. The curative track focusses on cure, whilst the palliative track focusses on the quality of life. Button et al. suggested an alternative disease trajectory model for haematological malignancies.⁵⁸ This model assumes that all patients with a haematological malignancy, no matter if the treatment is of curative or palliative intent, need a palliative care approach.



Fig. 1. Modified model of palliative care for people with a haematological malignancy. From Button E, Bolton M, Chan RJ, Chambers S, Butler J, Yates P. A palliative care model and conceptual approach suited to clinical malignant haematology. Palliative Medicine. 2019;33(5):483-485. doi:10.1177/0269216318824489⁵⁸

The two-track approach in **Chapter 4** suggests a continuous evaluation of the goals of care. The cues for proactive care that were collected in **Chapters 3 and 4** can also be used for reviewing if the care delivered is concordant with the wishes of patients and family:

- Documentation of proactive care plans, which could be the following documents:
 - o Care plans for anticipating future symptoms or worsening symptoms;
 - o Care plans directed at informing the general practitioner;
 - o Care plans written by the palliative care consultation team;
 - o Referrals to the palliative care consultation team.
- Documentation of limitations on life-sustaining treatments (i.e., no resuscitation, no ventilation, no admission to the intensive care unit).

In summary, commonly used indicators for quality of end-of-life cancer care may not be applicable for patients with a haematological malignancy and do not explain what quality proactive palliative care entails. Goal concordance, which is part of the twotrack approach, can be an indicator of proactive palliative care and can be found in documentation in patient records.

2. Timely identification of palliative care needs: ED-triggered palliative care in patients with advanced cancer

In order to produce good results and avoid (future) problems (for definition, see 8.3.1), patients with palliative care needs should be timely identified. Early identification of palliative care needs can be difficult. Patients use acute healthcare resources more often in the last phase of life. Reasons for admission to an Emergency Department (ED) or hospital ward include serious symptom burden, not being able to manage at home, caregiver burden or lacking resources in the community setting.⁵⁹⁻⁶² Reaching out for acute medical help in the last phase of life may indicate disease progression and urgent palliative care needs. Because of this, events such as hospitalisations and ED-visits can function as triggers to consider if the patient can benefit from a palliative care approach.⁶³⁻⁶⁷ **Chapter 3** illustrates that ED-visits triggered initiation of discussions on and documentation of limitations on life-sustaining treatments in many of the patients visiting the ED. This thesis addresses two elements of ED-triggered palliative care needs; and estimation of the urgency of palliative care needs using the Surprise Question and poor physical performance.

In **Chapter 3**, risk factors for approaching death were identified to support identification of patients with advanced cancer visiting the ED: a diagnosis of lung cancer, neurologic deterioration, dyspnoea, hypercalcaemia, and jaundice. Together with triggers for

[•] Documentation of communication about the condition of the patient between a hospital clinician or palliative care specialist and the general practitioner of the patient. This communication could have occurred via a letter, telephone call or transfer notes.

palliative care derived from previous literature, these risk factors were integrated into a risk assessment tool.⁶⁸ The use of these triggers can support ED-clinicians in choosing actions leading to appropriate care for vulnerable patients. These actions include goalsof-care discussions, assessment of patients and family needs, referral to the preferred place with appropriate care and referral to specialist palliative care for complex problems.

Although risk assessment tools for approaching death can be useful in the ED-setting to organise appropriate care,⁶⁹⁻⁷¹ they are not suited for early identification of patients with palliative care needs. The study in **Chapter 5** demonstrated that the surprise question was a predictor for death within one year with a high sensitivity and positive predictive value, and can be used as a screening tool to early identify patients with palliative care needs in patients with advanced cancer visiting the ED. Specificity and c-statistic improve if poor functional status (ECOG 3 or 4) is added as a second step following the surprise question. A similar effect has been demonstrated in a study in elderly visiting the ED demonstrated that adding physician experience in working years and the PREDICT criteria, which are triggers for palliative care, to the one-year surprise question, improved the c-statistic for predicting one-year mortality.⁷²

The studies presented in **Chapters 3**, **4** and **5** have confirmed that ED-visits have a trigger effect. Limitations on life-sustaining treatments were often discussed after a patient visited the ED, apparently giving food for thought about what appropriate care is. The use of risk assessment tools, and combining the surprise question with functional status, can support ED-triggered, appropriate and goal-concordant palliative care.

8.3.4. Thinking ahead to be able to act before things happen

1. A two-track approach

A two-track approach assumes that a palliative care approach is concurrent with a curative care approach. Using scientific literature, including Button's model of care, and clinical experience of clinicians who care for patients with a haematological malignancy, an integrated care model has been developed that assumes that care for patients with a life-threatening illness should *not only hope for the best, but also prepare for the rest* (illustrated in **Chapter 4**, Fig. 1).^{58, 73-76} Goals-of-care discussions are an essential part of both tracks. Although **Chapter 4** focusses on the illness trajectory of patients with a haematological malignancy, it is probable that a two-track approach can also support patients with other serious illnesses that may still be curable. Examples are patients suffering from stroke or transient ischemic attacks, severe infections, or patients with organ failure waiting for transplantation.

8

Chapter 8

Timely initiation of a two-track approach is challenging. Known barriers to initiate palliative care conversations indicated by clinicians are, amongst others:^{77.79}

- Lack of time to conduct such conversations properly
- The patient's wishes and expectations are unknown
- Difficulties dealing with the needs of family
- Wishes and expectations of immigrant patients are unknown
- End-of-life conversations can trouble the clinician-patient relationship
- Feeling uncomfortable or unprepared to conduct such conversations.

It seems that training and gaining experience in palliative care conversations can facilitate initiation of those conversations and the use of a two-track approach. After implementation of the serious illness communication programme of Paladino et al., non-specialist palliative care clinicians were more aware of the necessity to initiate serious illness conversations early.⁸⁰ They initiated conversations earlier in the illness trajectory, and conversations were more holistic and more multidimensional. They said they started to think more proactively. This study proves that the implementation of person-centred communication methods raises awareness among clinicians on the value of timely conversations about palliative care, and that these discussions are initiated earlier.

2. Discussing future scenarios

From this thesis follows that clinicians should inform their patients to empower them and that clinicians have a professional responsibility to signal, explain, and document future scenarios. According to the Netherlands quality framework for palliative care, the discussion and reporting of current and future needs and wishes should be a standard part of proactive palliative care.¹⁹ The study in **Chapter 7** revealed that patients often want to know about the future and that they do not know what the future holds for them. They are reliant on their clinicians to provide them this information in order to regain a sense of control over their care. A two-track approach demands a proactive attitude from the clinician. However, Slort et al. found that future scenarios were not anticipated by general practitioners during end-of-life conversations.⁸¹ Flierman et al. demonstrated that hospital clinicians sometimes wait for patients and family to express their wishes and preferences for the future, and do not initiate the topic themselves.⁸²

Some clinicians in the study in **Chapter 7** mentioned that the question prompt list may support them in explaining future scenarios to patients and family. Patient-reported outcome measurements (PROMs), such as the Edmonton Symptom Assessment System (ESAS), can support communication between patients, their family, and clinicians.⁸³ Brooks et al. demonstrated that the ESAS helped them to identify and set priorities for treatment, and guided the conversation with the clinician.⁸⁴ Handing out a palliative

care PROM to patients and family can therefore be considered as a proactive action. In **Chapter 6**, the Leiden Guide on Palliative care, which includes the Utrecht Symptom Diary (a Dutch adaptation of the ESAS) and a palliative care question prompt list, was used for data collection. The study in **Chapter 6** provides insight into which the symptoms patients and family have questions about. It seems that patients and family use the question prompt list to indicate about which symptoms they worry, such as shortness of breath and pain. It is known that symptoms that patients prioritise are not necessarily the symptoms that cause the highest burden.⁸⁵ This demonstrates that discussions about future scenarios can be tailored to the patient and family by using specific PROMs for assessment of symptom burden and information needs. **Chapter 6** demonstrates that the information needs of patients and family can differ by the phase of their illness. For example, patients whose treatment was mainly aimed at symptom management had more often information needs about managing care at home.

Discussing future scenarios is indicated and warranted: patients and family want and need to be informed about the future. The addition of the question prompt list to a symptom assessment scale supports clinicians to also ask about the worries of patients and family and to explain future scenarios.

8.4. RECOMMENDATIONS FOR FUTURE RESEARCH

Recommendations for future research following from this thesis will be addressed in three paragraphs:

- 1. Promoting knowledge and self-initiation in clinicians, and patients and family
- 2. Intention to produce good results and avoid (future) problems
- 3. Thinking ahead to be able to act before things happen

8.4.1 Promoting knowledge and self-initiation in clinicians, and patients and family

1. Good education for undergraduate medical students

Internationally, assessments of undergraduate medical education have resulted in implementation of end-of-life care courses or internships in the formal medical curricula. In the United Kingdom, for example, the assessment from 1983 revealed that four medical faculties did not pay any attention to end-of-life care.⁸⁶ Thereafter, the British medical curricula were changed in 1994 and 2002 and are regularly evaluated ever since.⁸⁷⁻⁸⁹ Switzerland and Germany made similar changes.⁹⁰⁻⁹² The study described in **Chapter 2** and the studies by Pieters et al. can be considered as baseline measurements of the

integration of end-of-life care and palliative care in the Dutch medical curricula.^{27, 29} To evaluate the effect of curricular interventions, it is preferable to measure the actual presence, knowledge, perceived importance and confidence again after implementation. An evaluation study on the presence of palliative care in the undergraduate medical curricula could be conducted by using a questionnaire or structured interviews among curricular programmers. Next, the effect of curricular changes should be measured using the methods of Chapter 2 or Pieters et al.²⁷ EPAs (see 8.3.2.1), and competencies that are now added to the updated blueprint on medical education, can be used as primary outcomes in a cross-sectional survey or interview study in medical students who are about to finish their undergraduate medical studies. In this way, actual competence and experienced competence can be measured in young medical doctors, and opportunities for improvement of undergraduate education on palliative care can be identified.

2. Patient and family empowerment

Patients and family indicated in **Chapter 7** that the question prompt list of the Leiden Guide on Palliative Care (LGP) increased their sense of control. Future research should be aimed at how much the sense of control of patients and family increases after using the LGP in terms of feeling empowered to manage their illness. Self-management can be considered as the last station in person-centred care, according to Pulverenti et al.:



Fig. 2. Person-centred care improves patient empowerment, which improves self-management in patients and family. Adapted from Pulvirenti et al.³⁴

Lorig defined self-management as "*learning and practicing the skills necessary to carry on an active and emotionally satisfying life in the face of a chronic condition*".⁹³ The main goal of self-management interventions is supporting the patient to live and achieve those outcomes he or she wishes. Dineen-Griffin et al. created a model that supports clinicians in conducting self-management interventions during consultations in a person-centred way. Essential elements of this model are:

- Provision of effective self-management support
- Theoretical base to the intervention
- Face-to-face multicomponent intervention with primary care provider
- Tailored combination of self-management supporting strategies relevant to patient needs
- Ongoing follow-up (face-to-face, telephone)

Further research should investigate whether appropriate use of a question prompt list, according to the model by Dineen-Griffin et al., not just increases patient empowerment, but also self-management. A study on improving self-management by using a question prompt list can have a randomised controlled trial design, in which one group of patients receives and discusses a question prompt list, and the second groups receives a 'standard' palliative care consultation or conversation without a question prompt list. Both groups receive person-centred care according to Dineen-Griffin's essential elements. The primary outcome would be patients and family's self-management. In this way, the value of a question prompt list on self-management of patients and family in the context of person-centred care can be determined.

A review by Wakefield et al. suggested that a patient satisfaction questionnaire may be a valuable addition to a question prompt list.⁴⁰ Further research can include the addition of patient satisfaction to the LGP, and be aimed at if this addition can improve personcentred care, patient empowerment and self-management. A study on an addition to the LGP could be an evaluation study including semi-structured interviews or questionnaires among patients and family.

Clinicians indicated in **Chapter 7** that the LGP can support palliative care consultations. However, for purposes to study whether non-specialists would support the use of the LGP, these clinicians had not used the LGP before. In advance of implementation of the LGP in several non-specialist settings, first a pilot study should be conducted among, for example, general practitioners to study the feasibility of using the LGP in their clinical practice. This pilot study can have a participatory action research (PAR) design. Using PAR, a working method can be developed using quantitative and qualitative research methodologies while improving care activities.⁹⁴ PAR includes the execution of action cycles, during which the working method can be refined, tested, and evaluated. PAR supports developing a working method in specific settings, which may be helpful in the complexity of care settings. Next, a cluster randomised trial in general practices can demonstrate whether the LGP improves 1) patient outcomes, such as quality of life, satisfaction, information needs and symptom burden, and 2) quality of palliative care and end-of-life care, using outcomes like quality indicators and goal concordance.

The LGP is mostly studied in patients with cancer. Further research involving the LGP should also include patients with non-cancer diseases to evaluate whether use of the LGP is suited in patients with a non-cancer diagnosis. This includes repeating the study presented in **Chapter 6**, which provides an oversight of symptom burden an information needs of patients of a palliative care consultation team of an academic medical centre, in another cohort of a palliative care consultation team. In this way, insight about symptom burden

8

and information needs of patients with other illness trajectories can be provided. This study design can also be used to study symptom burden and information needs in patients in care settings other than the hospital, such as in general practice and in nursing homes.

The research suggestions described above concern the LGP, but also apply to palliative care question prompt lists in general. To support future research on question prompt lists in palliative care, McDarby et al. recently composed a research agenda concerning the use of question prompt lists in outpatient palliative care.⁹⁵ The authors state that more knowledge and insights are needed on five questions:

- which patient populations with palliative care needs will use and benefit from using a question prompt list?
- what is the right timing of providing a question prompt list and how could a question prompt list be provided best in the context of outpatient palliative care consultations?
- which patient and family outcomes are addressed by using a question prompt list in outpatient palliative care, and how are these outcomes addressed?
- which characteristics of a question prompt list on palliative care contribute to the best benefits to patients and families in outpatient palliative care?
- what are the potential benefits of using a question prompt list for palliative care clinicians and healthcare organisations?

It is possible that the answers to each research question differ for the variety of question prompt lists, care settings and patient populations. It is important to study which situations demand certain kinds of question prompt lists. Additionally, it seems plausible that the same research questions hold for the use of question prompt lists inpatient palliative care, and that the answers may differ from outpatient palliative care.

8.4.2 Intention to produce good results and avoid (future) problems

1. Quality indicators for palliative care in patients with a haematological malignancy

In paragraph 8.3.3.1 is suggested that commonly used indicators for the quality of end-oflife care may not reflect how haematologists think what quality of end-of-life care entails for their patients. To develop a list of quality indicators for end-of-life care in patients with a haematological malignancy, a study with a Delphi technique can be useful. In such a study, haematologists with an interest for palliative care or a specialisation in palliative care can be asked to make a list of priorities regarding quality end-of-life care. Such a list of indicators of quality end-of-life care in haematology can subsequently be used in an evaluation study to find opportunities for improvement of proactive palliative care. A commonly used method for the evaluation of end-of-life care is a mortality follow-back design. The follow-back period can be three months for evaluating the terminal phase

General discussion

of the illness trajectory, or six months or one year for more upstream evaluation of the palliative phase of the illness trajectory.¹⁹

Early palliative care is preferable to timely prepare patients with a (potentially) lifethreatening illness and their family. However, a proactive palliative care approach can also be initiated later in the illness trajectory, which is especially imaginable in patients with illness trajectories that are difficult to predict. Kripp et al. have constructed a prediction score for approaching death consisting of low performance status (ECOG score >2), low platelet count (<90×10-9/L), opioid treatment for pain (WHO level 3), high plasma LDH (>248U/L) and low plasma albumin (<30g/L).⁹⁶ Using these predictors, three risk groups can be identified: low risk (presence of no or one predictor; median survival of 440 days); intermediate risk (two or three predictors; median survival of 63 days); and high risk (median survival of 10 days). The Kripp prediction score can support identification of patients with palliative care needs and differentiate in urgency of their needs. However, this study was conducted in patients who were admitted to a palliative care unit, suggesting that these patients were already identified as having palliative care needs. Validation of the Kripp prediction score in a prospective cohort study in patients admitted to the haematology ward is needed to know whether the Kripp prediction score can be used to identify patients with approaching death.

2. Timely identification of palliative care needs: ED-triggered palliative care in patients with advanced cancer

An ED-visit can be a trigger to initiate proactive palliative care.⁶⁵ In **Chapter 3** and **Chapter 4**, it seems that the ED-visit triggered discussions about limitations on lifesustaining treatments. A prospective cohort study in patients identified at the ED using the surprise question can provide insight into 1) whether ED-visits increase the number of documentation of goals-of-care discussions, and 2) goal concordance of care following the ED-visit.

The trigger card presented in **Chapter 3** included risk factors for approaching death in patients with advanced cancer. To validate the trigger card and to identify other predictors for approaching death, a prospective cohort study can be conducted in patients with a diagnosis of advanced cancer (without curative treatment options or with curative treatments with a substantial risk of mortality). 'Death' can be a candidate outcome measure to identify predictors for patients in the dying phase. However, 'death within three months' could be more clinically relevant in organising appropriate palliative terminal care at the place patients and their family would prefer.

Chapter 5 demonstrated that patients with advanced cancer, about whom the clinician would not be surprised if they died within one year, and who have a poor physical

performance (SQ plus ECOG 3-4), can be used to differentiate in the urgency of their needs. Research should be aimed at the question whether the use of SQ plus ECOG, as described in Table 3 in **Chapter 5**, improves patient outcomes such as quality of life and symptom burden. This could be studied in an ED using SQ plus ECOG to identify patients with palliative care needs in a prospective follow-up study design including questionnaires distributed to patients after 1, 3 and 6 months, and after 1 year. After identification in the ED, a multidimensional screening of problems should take place, for example by referring to the palliative care consultation team, or by screening by ED-clinicians using the 5-SPEED (see 8.5.2.2). As part of a cluster randomised trial, these outcomes can be compared with an ED in which standard care is provided. Also, SQ plus ECOG could be used to provide insight into the outcomes of ED-triggered palliative care as a care model. The outcomes of such a study can be a multidimensional approach to symptoms and the quality of palliative or end-of-life care including goal concordance.

The study in **Chapter 5** was conducted in patients who were already admitted to the ED. SQ plus ECOG could also provide insight if asked to the admitting clinician before admission to the ED. It could trigger goals-of-care discussions before ED-admission, which can contribute to appropriate care, with or without admission. This could be studied using a prospective cohort study design, in which the SQ plus ECOG could be asked to the ordering clinician during the triage process if it concerns a patient with advanced cancer. The primary outcome can be goal concordance of care, measured from data collected from patient records, for example by using the cues for documentation of proactive palliative care from 8.3.3.1.

It is known that conducting palliative care implementation research in the ED can be challenging. George et al. found that their screening tool was considered feasible in clinical practice by 70% of the ED-clinicians; however, although the 78% of patients who were found by screening as having unmet palliative care needs, only 26% was referred to palliative care.^{97, 98} Successful implementation of a multidimensional screening intervention in the ED therefore depends on the quality of the implementation process. In Leiden University Medical Center's ED, the acutely presenting older patient (APOP) screener was carefully introduced in the workflow of this ED. The implementation strategy, which was part of a plan-do-study-act study, included acquiring information from clinicians and patients, adaptation of the screener where appropriate, development of standard operation procedures, integration into electronic patient records and an educational intervention, which all occurred pre-implementation. Already two months after implementation, 31% vs 21% (p=0.0002) received a comprehensive geriatric assessment.99 This demonstrates that a plan-do-study-act strategy, which is part of participatory action research, can be an effective way to implement and study palliative care interventions in the FD.

8.4.3 Thinking ahead to be able to act before thing happen

1. A two-track approach

A two-track approach as described in **Chapter 4** and in paragraph 8.3.4.1 of this chapter can improve proactive care by concurrent tracks of curative care and palliative care. To develop a two-track model of care in a haematology department, a participatory action research (PAR) approach can be used. The development of a two-track model of care as a working method using PAR not only provides insight into how the model works in clinical practice, but also aims at implementing the model into daily care. It is therefore essential to make arrangements with all involved stakeholders, including the generalist palliative care responsibilities of haematology clinicians and the specialist palliative care responsibilities of palliative care consultants. These responsibilities are described by Henderson et al.²⁵ Additionally, haematology clinicians should be educated about generalist palliative care in the context of their responsibilities in the palliative care track.

The description of the two-track care model as a working method and how it can be implemented can support hospital wards to start working with a two-track care model. Using the quality indicators of paragraph 8.3.3.1, the quality of care can be evaluated before and after implementation of the two-track care model. The implementation should ideally be part of a cluster randomised trial, in which haematology wards of multiple hospitals are randomised to either standard care or the two-track care model. The primary outcome should be the quality of life, and potential secondary outcomes are multidimensional symptom burden, satisfaction with care and goal concordance of care.

The haematology palliative care model as described by Button (paragraph 8.3.3.1) assumes that all patients with a haematological malignancy need a palliative care approach because of the high mortality risk due to either the illness or the treatment.⁵⁸ The Button model is probably also appropriate in other illnesses that have a similar trajectory, for example aggressive but treatable carcinomas such as advanced stage melanomas, or osteosarcomas and pancreatic carcinomas that may be treated with surgery but may also have metastasized. Also non-cancer patients may have a 'haematology-like' trajectory, for example patients with serious infections acquiring intensive treatment and patients with organ failure waiting for transplantation. The aforementioned two-track care model could also be researched using PAR to develop and implement a working method for concurrent palliative care for these patients.

A two-track approach demands a proactive attitude in discussing palliative care needs. In **Chapters 3, 4 and 5**, cues for proactive palliative care were used to study the end-of-life trajectories of patients with advanced cancer or a haematological malignancy visiting the ED (see also paragraph 8.3.3.1). This list should be evaluated and complemented in

a study among experts, for example by using a Delphi technique. The list with cues for proactive palliative care can be used in evaluation studies on the proactivity of palliative care, for example in a cohort study in patients identified as having palliative care needs.

2. Discussing future scenarios

The discussion of future scenarios is part of a two-track approach. **Chapter 6** has demonstrated that patients and family have information needs about symptoms and situations that are not actual yet. In **Chapter 7**, non-specialist palliative care clinicians indicated that the Leiden Guide on Palliative Care (LGP) included items that can support discussing and explaining future scenarios to patients and their family during palliative care conversations. To study if non-specialist clinicians feel supported in discussing future scenarios by using the LGP, a study can be conducted in two groups of clinicians: 1) those trained to use the LGP, and 2) those who were not trained to use the LGP. Both groups can be asked by using interviews or a questionnaire if they discuss future scenarios, and if they feel comfortable doing it. Also, for example one and six months after implementation, patient records can be searched for if discussed future scenarios are documented, and if the number of discussions is increased in the group that was trained in using the LGP.

8.5. RECOMMENDATIONS FOR EDUCATION, CLINICAL PRACTICE, AND POLICY

Recommendations for education, clinical practice, and policy following from this thesis will be addressed in three paragraphs:

- 1. Promoting knowledge and self-initiation in clinicians, and patients and family
- 2. Intention to produce good results and avoid (future) problems
- 3. Thinking ahead to be able to act before things happen

8.5.1 Promoting knowledge and self-initiation in clinicians, and patients and family

1. Good education for undergraduate medical students

Following the results from **Chapter 2**, addition of the five essential domains of end-of-life care education to the Dutch national blueprint on medical education was recommended. The study was one of the many efforts to improve undergraduate medical education on palliative care, including the PASEMECO and the O2PZ projects. With the revised blueprint published in 2020, generalist palliative care has become a compulsory part of the formal medical curricula in the Netherlands.

The PASEMECO toolbox and integral palliative care course are examples that can be useful for curriculum programmers and teachers to optimise the curricula.^{27, 100} Among other aims (see 8.3.2.1), O2PZ is committed to supporting curriculum programmers in integrating palliative care into the packed medical curricula. In this way, palliative care education can be integrated in a way that suits the local curricula, and in coordination with curriculum programmers. It may be difficult to persuade curriculum programmers to make space for palliative care education. A suggestion to discuss with curriculum programmers is that all medical doctors should have generalist palliative care competences, because the palliative care demand will increase (see **Chapter 1**), and these generalist competences are described in the Quality framework on palliative care.^{19, 20, 25}

Measuring and evaluating the quality of undergraduate medical education can result in better anchoring of palliative care in the curricula (see 8.4.1.1). Therefore, regular evaluation of undergraduate medical curricula is not only needed for research purposes, but also to guard the preparation of medical students for their future work, including generalist palliative care competences.

2. Patient and family empowerment

The Leiden Guide on Palliative Care (LGP), a palliative care conversation guide, combines assessment of symptom burden with assessment of information needs. Using feedback of patients, family and clinicians in **Chapter 7**, an instruction leaflet was developed (Supplement 7 of Chapter 7) in order to support clinicians in using the question prompt list appropriately and optimally. This instruction leaflet can be used in training and education about the empowerment of patients and family during individual conversations about palliative care. Good training in using the LGP is necessary since use of the question prompt list may evoke strong emotions in patients and family (**Chapter 7**), even though they expect to talk about difficult questions cohering to the phase of their illness.

Chapter 6 demonstrated that symptom assessment alone is not sufficient to explore what topics patients worry about. It seems that patients and family want to know more about the future and how they can manage the illness themselves. From the results of this thesis follows that the LGP can be supportive in both assessing (information) needs, and providing person-centred advice appropriate for the phase of their illness. The use of a combination of a symptom assessment with a question prompt list in clinical practice is recommended.

A side outcome of this thesis is the uncertainty patients, family, lay people, but also clinicians, may experience about what palliative care actually entails. Appropriate and sufficient education of clinicians, patients and family is essential when reflecting on the core principles of proactive palliative care:

8

Core principles of proactive palliative care

- Promoting knowledge and self-initiation in clinicians, patients, and family
- Intention to produce good results and avoid (future) problems
- Thinking ahead to be able to act before things happen

This thesis only included studies relevant to individual palliative care consultations. To really improve proactivity in the triad patients, family and clinicians, community-based interventions should be initiated. When this thesis was being written, the SIRE (Stichting Ideële Reclame) initiated nation-wide advertisements about talking about death on television, radio, and the internet. References to websites and other sources supporting people to get more knowledgeable about palliative care are essential nowadays to reach people. Indeed, first knowledge should be promoted before self-initiation, intention to produce good results and avoid (future) problems, and thinking ahead to be able to act before things happen, can be achieved.

8.5.2 Intention to produce good results and avoid (future) problems

1. Quality indicators for palliative care in patients with a haematological malignancy

The concept of palliative care is often explained using the model of Lynn and Adams and the model of Murray (**Chapter 1**). These models assume that the palliative phase includes a phase with disease-modifying treatment and/or symptom-directed treatment, terminal phase, and bereavement care. There are in general three illness trajectories, namely advanced cancer, organ failure and frailty, and the multidimensional needs of patients and family can change along the illness trajectory. The illness trajectory of patients with a haematological malignancy does not fit these models. The model proposed by Button et al. is more appropriate for patients with a haematological malignancy (paragraph 8.3.3.1).⁵⁸ This model can be applicable not only for patients with a haematological malignancy, but also for patients with an unpredictable illness trajectory with a substantial risk of death because of either the illness or the treatment. The Button model should be part of generalist palliative care education since it raises awareness for a concurrent palliative care track while undergoing curative treatment.

In quality improvement projects, haematology specific quality indicators should be used. In 8.4.2 is discussed how a list of validated haematology quality indicators can be derived. In addition, in aiming at a proactive and person-centred care, the outcome of a quality improvement project can be the goal concordance of treatments with the wishes of patients and family.

2. Timely identification of palliative care needs: ED-triggered palliative care in patients with advanced cancer

ED-visits can be a trigger for the initiation of palliative care and can improve the quality of life.^{64,66} Using the results in **Chapter 3**, and a report on care in the last phase of life in patients with advanced colon or lung cancer of by the Netherlands Care Institute (Zorginstituut Nederland), a trigger card was constructed that can support timely recognition of patients with advanced cancer at high risk of approaching death.⁶⁸ The trigger card supports taking directed actions in patients with urgent palliative care needs visiting the ED and can be implemented in the ED with help from a palliative care consultation team. Palliative care specialists should give instructions on how to use the trigger card, when to conduct goals-of-care discussions, and when to refer to specialist palliative care.

In **Chapter 4**, the predictive value of the surprise question (SQ) and the addition of other potential predictors for palliative care needs were studied. The SQ plus performance score of 3 or 4 according to the Eastern Cooperative Oncology Group (ECOG) can help differentiate in the urgency of palliative care needs in patients with advanced cancer. The clinical implications for using SQ plus ECOG 3-4 are summarised below (Table 1).

Patients with advanced cancer	Median survival	Actions		
Step 1: Surprise Question				
NS	3 months	Initiate symptom-management and focus care on quality of life		
Step 2: ECOG				
NS + ECOG 0-2; S	6 months	Initiate a two-track approach by complementing disease- modifying treatment with conversations about the potential end of life and the patient's wishes to prepare patients and family		
NS + ECOG 3-4	1 month	Initiate end-of-life care according to the patient's wishes as soon as possible		

Table 1. Differentiating in urgency of palliative care needs in patients with advanced cancer visiting the emergency department, using the Surprise Question and functional status (using the ECOG classification), and appropriate actions

Abbreviations: S: surprised; NS: not surprised; ECOG: Eastern Cooperative Oncology Group

After identification of patients who can benefit from a palliative care approach, they should be screened for symptoms or problems in a multidimensional way. George et al. developed a tool to screen for multidimensional palliative and end-of-life care needs in the ED, derived from a systematic review and shortened: 5-SPEED.^{101, 102} (5-SPEED):¹⁰³

- 1. pain management: how much are you suffering from pain?
- 2. home care: how much difficulty are you having getting your care needs met at home?
- 3. medication management: how much difficulty are you having with your medications?

- 4. psychological support: how much are you suffering from feeling overwhelmed?
- 5. goals of care: how much difficulty are you having getting medical care that fits with your goals?

Patients are instructed to score their answers to these questions on a 0-10 Likert scale. The 5-SPEED is validated for use in patients with cancer visiting the ED.¹⁰¹

8.5.3 Thinking ahead to be able to act before things happen

1. A two-track approach

In paragraph 8.4.2 is proposed that the Button illness trajectory model for patients with a haematological malignancy should be taught together with the more common used illness trajectories for patients with advanced cancer, organ failure and frailty. A twotrack approach is a concrete application of the Button model. Education about the twotrack approach in clinical practice can support non-specialist palliative care clinicians to discuss curative and palliative approaches to care with patients and their families. It is also possible to make arrangements with specialist palliative care clinicians, for example conducting the two-track approach together. In this case, the attending clinician can keep their role in the curative track, and the specialist palliative care clinician could be responsible for the palliative track. This may also help patients and family to bring up issues palliative care needs without having to worry that it will influence their curative treatment. It is important to note in education that initiating a palliative track is not only appropriate in patients who will certainly die soon; rather, patients with a possible lifethreatening illness can have palliative care needs while undergoing intensive curative treatment. A two-track approach should be advertised as an approach that can answer the needs of these vulnerable patients.

2. Discussing future scenarios

Discussing future scenarios with patients and family is essential in providing proactive palliative care. However, current medical practice is mainly focused on the treatment of diseases (i.e., a more reactive approach). More attention should be invested in education and bedside teaching on prevention of unwanted outcomes, especially for clinicians providing generalist palliative care. A helpful model for initiating discussions about possible unfavourable outcomes in life-threatening illnesses and potentially curative treatments is the Goals of Car(e) model. Hui and Bruera used this model in an oncology clinic as part of integrated care. The Goals of Car(e) model uses a car as an analogy for the discussion of goals of care; not only the aspects of curative treatment, but also the aspects of supportive, palliative and hospice care should be discussed in order to prepare patients and family for a safe and pleasant road trip.¹⁰⁴

Prevention can occur during individual consultations, but also on meso- and macro-level. The discussion about the prevention of unwanted outcomes can be initiated upfront, for example in healthy individuals: how do you wish your dying phase would look like, and what does absolutely not match how you want to die? Discussions about future scenarios can however take place at any time during an illness trajectory, since at any time unwanted outcomes can be prevented if discussed with patients and their family.

CONCLUSION

Proactive palliative care includes three core principles: promotion of knowledge and selfinitiation in the triad of clinician, patient, and family; the intention to produce good results and avoid (future) problems; thinking ahead to be able to act before things happen. This chapter outlined which opportunities for improving proactive palliative care are provided by this thesis: the need for quality palliative care education for clinicians, improvement of patient and family empowerment, quality indicators for end-of-life care, timely identification of palliative care needs, and a two-track approach for timely discussions about palliative care and future scenarios.

8

REFERENCES

- 1. van der Baan FH, Koldenhof JJ, de Nijs EJ, et al. Validation of the Dutch version of the Edmonton Symptom Assessment System. *Cancer Med* 2020 2020/07/10. DOI: 10.1002/ cam4.3253.
- 2. Clayton J, Butow P, Tattersall M, et al. Asking questions can help: development and preliminary evaluation of a question prompt list for palliative care patients. *Br J Cancer* 2003; 89: 2069-2077. 2003/12/04. DOI: 10.1038/sj.bjc.6601380.
- 3. Parker SKB, U.K.; Strauss, K. Making Things Happen: A Model of Proactive Motivation. *Journal of Management* 2010; 36: 827-856. DOI: https://doi.org/10.1177%2F0149206310363732.
- 4. Kahana E and Kahana B. Patient proactivity enhancing doctor-patient-family communication in cancer prevention and care among the aged. *Patient Educ Couns* 2003; 50: 67-73. 2003/05/28. DOI: 10.1016/s0738-3991(03)00083-1.
- Henrike HW and Schultz C. The impact of health care professionals' service orientation on patients' innovative behavior. *Health Care Manage Rev* 2014; 39: 329-339. 2013/07/31. DOI: 10.1097/HMR.0b013e31829d534c.
- Tuohy D and Cooney A. Older Women's Experiences of Aging and Health: An Interpretive Phenomenological Study. *Gerontol Geriatr Med* 2019; 5: 2333721419834308. 2019/03/20. DOI: 10.1177/2333721419834308.
- Demes JAE, Nickerson N, Farand L, et al. What are the characteristics of the champion that influence the implementation of quality improvement programs? *Eval Program Plann* 2020; 80: 101795. 2020/02/23. DOI: 10.1016/j.evalprogplan.2020.101795.
- 8. Allen M and Watts T. Promoting health and wellbeing at the end of life: the contribution of care pathways. *Int J Palliat Nurs* 2012; 18: 348-354. 2012/08/14. DOI: 10.12968/ ijpn.2012.18.7.348.
- **9.** Murray SA, Kendall M, Boyd K, et al. Illness trajectories and palliative care. *BMJ* 2005; 330: 1007-1011. 2005/04/30. DOI: 10.1136/bmj.330.7498.1007.
- **10.** Murray SA, Kendall M, Mitchell G, et al. Palliative care from diagnosis to death. *BMJ* 2017; 356: j878. 2017/03/01. DOI: 10.1136/bmj.j878.
- **11.** Duenk RG, Verhagen C, Bronkhorst EM, et al. Development of the ProPal-COPD tool to identify patients with COPD for proactive palliative care. *Int J Chron Obstruct Pulmon Dis* 2017; 12: 2121-2128. 2017/08/10. DOI: 10.2147/COPD.S140037.
- 12. Duenk RG, Verhagen C, Bronkhorst EM, et al. Proactive palliative care for patients with COPD (PROLONG): a pragmatic cluster controlled trial. *Int J Chron Obstruct Pulmon Dis* 2017; 12: 2795-2806. 2017/10/17. DOI: 10.2147/COPD.S141974.
- **13.** Hackett J, Ziegler L, Godfrey M, et al. Primary palliative care team perspectives on coordinating and managing people with advanced cancer in the community: a qualitative study. *BMC Fam Pract* 2018; 19: 177. 2018/11/22. DOI: 10.1186/s12875-018-0861-z.
- 14. Liyanage T, Mitchell G and Senior H. Identifying palliative care needs in residential care. *Aust J Prim Health* 2018; 24: 524-529. 2018/11/14. DOI: 10.1071/PY17168.
- **15.** Johnson CE, McVey P, Rhee JJ, et al. General practice palliative care: patient and carer expectations, advance care plans and place of death-a systematic review. *BMJ Support Palliat Care* 2018 2018/07/27. DOI: 10.1136/bmjspcare-2018-001549.
- **16.** Oosterink JJ, Oosterveld-Vlug MG, Glaudemans JJ, et al. Interprofessional communication between oncologic specialists and general practitioners on end-of-life issues needs improvement. *Fam Pract* 2016; 33: 727-732. 2016/09/03. DOI: 10.1093/fampra/cmw064.
- **17.** Thoonsen B, Groot M, Verhagen S, et al. Timely identification of palliative patients and anticipatory care planning by GPs: practical application of tools and a training programme. *BMC Palliat Care* 2016; 15: 39. 2016/04/06. DOI: 10.1186/s12904-016-0112-9.

- Verschuur EML, Groot MM and van der Sande R. Nurses' perceptions of proactive palliative care: a Dutch focus group study. *Int J Palliat Nurs* 2014; 20: 241-245. 2014/05/24. DOI: 10.12968/ijpn.2014.20.5.241.
- Integraal Kankercentrum Nederland (IKNL). Netherlands Quality Framework for Palliative Care, www.palliaweb.nl/publicaties/netherlands-quality-framework-for-palliative-care (2017).
- 20. Boddaert MS, Douma J, Dijxhoorn AQ, et al. Development of a national quality framework for palliative care in a mixed generalist and specialist care model: A whole-sector approach and a modified Delphi technique. *PLoS One* 2022; 17: e0265726. 2022/03/24. DOI: 10.1371/ journal.pone.0265726.
- 21. World Health Organization Definition of Palliative Care, http://www.who.int/cancer/ palliative/definition/en/ (accessed 05-10-2018).
- **22.** Just because we can, doesn't mean we should. Appropriate end-of-life care. 2015 2015. Utrecht, the Netherlands: Royal Dutch Medical Association.
- 23. Bolt EEP, H.R.W.; Willems, D.L.; Onwuteaka-Philipsen, B.D. Passende zorg in de laatste levensfase. *Nederlands Tijdschrift voor Geneeskunde* 2017; 161.
- 24. de Bruin J, Verhoef M, Slaets JP, et al. Toekomstige artsen beter voorbereiden op levenseindezorg. *Nederlands Tijdschrift voor Geneeskunde* 2020; 164: 1-4. Perspectief.
- **25.** Henderson JD, Boyle A, Herx L, et al. Staffing a Specialist Palliative Care Service, a Team-Based Approach: Expert Consensus White Paper. *J Palliat Med* 2019; 22: 1318-1323. 2019/07/28. DOI: 10.1089/jpm.2019.0314.
- **26.** Rosa WE, Ferrell BR and Mason DJ. Integration of Palliative Care Into All Serious Illness Care as A Human Right. *JAMA Health Forum* 2021; 2: e211099-e211099. DOI: 10.1001/ jamahealthforum.2021.1099.
- 27. Pieters J, Dolmans D, Verstegen DML, et al. Palliative care education in the undergraduate medical curricula: students' views on the importance of, their confidence in, and knowledge of palliative care. *BMC Palliat Care* 2019; 18: 72. 2019/08/29. DOI: 10.1186/s12904-019-0458-x.
- **28.** Hoes A, Hogendoorn P, Joels M, et al. *Raamplan Artsopleiding 2020.* 2020. Houten: NFU Nederlandse Federatie van Universitair Medische Centra.
- 29. Pieters J, Dolmans D, van den Beuken-van Everdingen MHJ, et al. A National, Palliative Care Competency Framework for Undergraduate Medical Curricula. *Int J Environ Res Public Health* 2020; 17 2020/04/05. DOI: 10.3390/ijerph17072396.
- **30.** Palliaweb. Onderwijsmaterialen, https://palliaweb.nl/onderwijsmaterialen (2022, accessed 05-04-2022).
- **31.** O2PZ. Over O2PZ, https://www.o2pz.nl/over+o2pz/over+opz/default.aspx (accessed 05-04-2022).
- O2PZ. Onderwijsraamwerk palliatieve zorg 2.0, https://www.o2pz.nl/actueel/ publicaties/1761450.aspx?t=Onderwijsraamwerk-palliatieve-zorg-20 (2020).
- Little P, Everitt H, Williamson I, et al. Preferences of patients for patient centred approach to consultation in primary care: observational study. *BMJ* 2001; 322: 468-472. 2001/02/27. DOI: 10.1136/bmj.322.7284.468.
- Pulvirenti M, McMillan J and Lawn S. Empowerment, patient centred care and selfmanagement. *Health Expect* 2014; 17: 303-310. 2012/01/04. DOI: 10.1111/j.1369-7625.2011.00757.x.
- **35.** Rathert C, Wyrwich MD and Boren SA. Patient-centered care and outcomes: a systematic review of the literature. *Med Care Res Rev* 2013; 70: 351-379. 2012/11/22. DOI: 10.1177/1077558712465774.
- **36.** Bauman AE, Fardy HJ and Harris PG. Getting it right: why bother with patient-centred care? *Med J Aust* 2003; 179: 253-256. 2003/08/20. DOI: 10.5694/j.1326-5377.2003.tb05532.x.

- **37.** McWilliam CL. Patients, persons or partners? Involving those with chronic disease in their care. *Chronic Illn* 2009; 5: 277-292. 2009/11/26. DOI: 10.1177/1742395309349315.
- **38.** Finfgeld DL. Empowerment of individuals with enduring mental health problems: results from concept analyses and qualitative investigations. *ANS Adv Nurs Sci* 2004; 27: 44-52. 2004/03/19. DOI: 10.1097/00012272-200401000-00006.
- **39.** Rolland JS. Cancer and the family: an integrative model. *Cancer* 2005; 104: 2584-2595. 2005/11/05. DOI: 10.1002/cncr.21489.
- **40.** Wakefield D, Bayly J, Selman LE, et al. Patient empowerment, what does it mean for adults in the advanced stages of a life-limiting illness: A systematic review using critical interpretive synthesis. *Palliat Med* 2018; 32: 1288-1304. 2018/06/30. DOI: 10.1177/0269216318783919.
- Walczak A, Butow PN, Davidson PM, et al. Patient perspectives regarding communication about prognosis and end-of-life issues: how can it be optimised? *Patient Educ Couns* 2013; 90: 307-314. 2011/09/17. DOI: 10.1016/j.pec.2011.08.009.
- **42.** Zimmermann C, Swami N, Krzyzanowska M, et al. Early palliative care for patients with advanced cancer: a cluster-randomised controlled trial. *Lancet* 2014; 383: 1721-1730. 2014/02/25. DOI: 10.1016/S0140-6736(13)62416-2.
- **43.** Haun MW, Estel S, Rucker G, et al. Early palliative care for adults with advanced cancer. *Cochrane Database Syst Rev* 2017; 6: CD011129. 2017/06/13. DOI: 10.1002/14651858. CD011129.pub2.
- **44.** Vanbutsele G, Pardon K, Van Belle S, et al. Effect of early and systematic integration of palliative care in patients with advanced cancer: a randomised controlled trial. *Lancet Oncol* 2018; 19: 394-404. 2018/02/07. DOI: 10.1016/S1470-2045(18)30060-3.
- **45.** Vanbutsele G, Van Belle S, Surmont V, et al. The effect of early and systematic integration of palliative care in oncology on quality of life and health care use near the end of life: A randomised controlled trial. *Eur J Cancer* 2020; 124: 186-193. 2019/12/10. DOI: 10.1016/j. ejca.2019.11.009.
- **46.** Bakitas MA, Tosteson TD, Li Z, et al. Early Versus Delayed Initiation of Concurrent Palliative Oncology Care: Patient Outcomes in the ENABLE III Randomized Controlled Trial. *J Clin Oncol* 2015; 33: 1438-1445. 2015/03/25. DOI: 10.1200/JCO.2014.58.6362.
- **47.** Dionne-Odom JN, Azuero A, Lyons KD, et al. Benefits of Early Versus Delayed Palliative Care to Informal Family Caregivers of Patients With Advanced Cancer: Outcomes From the ENABLE III Randomized Controlled Trial. *J Clin Oncol* 2015; 33: 1446-1452. 2015/03/25. DOI: 10.1200/JCO.2014.58.7824.
- **48.** McDonald J, Swami N, Hannon B, et al. Impact of early palliative care on caregivers of patients with advanced cancer: cluster randomised trial. *Ann Oncol* 2017; 28: 163-168. 2016/10/01. DOI: 10.1093/annonc/mdw438.
- **49.** Kavalieratos D, Corbelli J, Zhang D, et al. Association Between Palliative Care and Patient and Caregiver Outcomes: A Systematic Review and Meta-analysis. *JAMA* 2016; 316: 2104-2114. 2016/11/29. DOI: 10.1001/jama.2016.16840.
- **50.** Boddaert MS, Pereira C, Adema J, et al. Inappropriate end-of-life cancer care in a generalist and specialist palliative care model: a nationwide retrospective populationbased observational study. *BMJ Support Palliat Care* 2020 2020/12/24. DOI: 10.1136/ bmjspcare-2020-002302.
- **51.** De Korte-Verhoef MC, Pasman HR, Schweitzer BP, et al. Reasons for hospitalisation at the end of life: differences between cancer and non-cancer patients. *Support Care Cancer* 2014; 22: 645-652. 2013/10/23. DOI: 10.1007/s00520-013-2019-9.
- **52.** De Korte-Verhoef MC, Pasman HR, Schweitzer BP, et al. General practitioners' perspectives on the avoidability of hospitalizations at the end of life: A mixed-method study. *Palliat Med* 2014; 28: 949-958. 2014/04/04. DOI: 10.1177/0269216314528742.

- **53.** Earle CC, Park ER, Lai B, et al. Identifying potential indicators of the quality of end-of-life cancer care from administrative data. *J Clin Oncol* 2003; 21: 1133-1138. 2003/03/15. DOI: 10.1200/JCO.2003.03.059.
- 54. Earle CC, Neville BA, Landrum MB, et al. Evaluating claims-based indicators of the intensity of end-of-life cancer care. *Int J Qual Health Care* 2005; 17: 505-509. 2005/06/30. DOI: 10.1093/intqhc/mzi061.
- 55. Odejide OO, Salas Coronado DY, Watts CD, et al. End-of-life care for blood cancers: a series of focus groups with hematologic oncologists. *J Oncol Pract* 2014; 10: e396-403. 2014/10/09. DOI: 10.1200/JOP.2014.001537.
- **56.** Odejide OO and Steensma DP. Patients with haematological malignancies should not have to choose between transfusions and hospice care. *Lancet Haematol* 2020; 7: e418-e424. 2020/05/04. DOI: 10.1016/S2352-3026(20)30042-9.
- 57. Odejide OO, Cronin AM, Condron NB, et al. Barriers to Quality End-of-Life Care for Patients With Blood Cancers. *J Clin Oncol* 2016; 34: 3126-3132. 2016/07/13. DOI: 10.1200/ JCO.2016.67.8177.
- Button E, Bolton M, Chan RJ, et al. A palliative care model and conceptual approach suited to clinical malignant haematology. *Palliat Med* 2019; 33: 483-485. 2019/01/29. DOI: 10.1177/0269216318824489.
- 59. Barbera L, Taylor C and Dudgeon D. Why do patients with cancer visit the emergency department near the end of life? CMAJ 2010; 182: 563-568. 2010/03/17. DOI: 10.1503/ cmaj.091187.
- **60.** Mayer DK, Travers D, Wyss A, et al. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. *J Clin Oncol* 2011; 29: 2683-2688. 2011/05/25. DOI: 10.1200/JCO.2010.34.2816.
- **61.** Mercadante S, Masedu F, Valenti M, et al. The characteristics of advanced cancer patients followed at home, but admitted to the hospital for the last days of life. *Intern Emerg Med* 2016; 11: 713-718. 2016/02/20. DOI: 10.1007/s11739-016-1402-1.
- **62.** Cooper E, Hutchinson A, Sheikh Z, et al. Palliative care in the emergency department: A systematic literature qualitative review and thematic synthesis. *Palliat Med* 2018: 269216318783920. 2018/07/22. DOI: 10.1177/0269216318783920.
- **63.** Grudzen CR, Stone SC and Morrison RS. The palliative care model for emergency department patients with advanced illness. *J Palliat Med* 2011; 14: 945-950. 2011/07/20. DOI: 10.1089/jpm.2011.0011.
- **64.** Kistler EA, Sean Morrison R, Richardson LD, et al. Emergency department-triggered palliative care in advanced cancer: proof of concept. *Acad Emerg Med* 2015; 22: 237-239. 2015/02/03. DOI: 10.1111/acem.12573.
- **65.** Kistler EA, Stevens E, Scott E, et al. Triggered Palliative Care Consults: A Systematic Review of Interventions for Hospitalized and Emergency Department Patients. *J Pain Symptom Manage* 2020 2020/02/18. DOI: 10.1016/j.jpainsymman.2020.02.001.
- **66.** Grudzen CR, Richardson LD, Johnson PN, et al. Emergency Department-Initiated Palliative Care in Advanced Cancer: A Randomized Clinical Trial. *JAMA Oncol* 2016 2016/01/16. DOI: 10.1001/jamaoncol.2015.5252.
- **67.** Mahony SO, Blank A, Simpson J, et al. Preliminary report of a palliative care and case management project in an emergency department for chronically ill elderly patients. *J Urban Health* 2008; 85: 443-451. 2008/03/26. DOI: 10.1007/s11524-008-9257-z.
- **68.** Zorginstituut Nederland. *Verbetersignalement: Zorg in de laatste levensfase bij mensen met darm- of longkanker.* 2017.
- **69.** Morita T, Tsunoda J, Inoue S, et al. The Palliative Prognostic Index: a scoring system for survival prediction of terminally ill cancer patients. *Support Care Cancer* 1999; 7: 128-133. 1999/05/21.

- **70.** Maltoni M, Nanni O, Pirovano M, et al. Successful validation of the palliative prognostic score in terminally ill cancer patients. Italian Multicenter Study Group on Palliative Care. *J Pain Symptom Manage* 1999; 17: 240-247. 1999/04/16.
- **71.** McMillan DC. An inflammation-based prognostic score and its role in the nutrition-based management of patients with cancer. *Proc Nutr Soc* 2008; 67: 257-262. 2008/05/03. DOI: 10.1017/S0029665108007131.
- 72. Ouchi K, Jambaulikar G, George NR, et al. The "Surprise Question" Asked of Emergency Physicians May Predict 12-Month Mortality among Older Emergency Department Patients. *J Palliat Med* 2018; 21: 236-240. 2017/08/29. DOI: 10.1089/jpm.2017.0192.
- **73.** Zimmermann C. Palliative care for patients with hematological malignancies: Time for a new model. *Leuk Res* 2016; 48: 78-79. 2016/08/09. DOI: 10.1016/j.leukres.2016.07.012.
- **74.** LeBlanc TW and El-Jawahri A. When and why should patients with hematologic malignancies see a palliative care specialist? *Hematology Am Soc Hematol Educ Program* 2015; 2015: 471-478. 2015/12/08. DOI: 10.1182/asheducation-2015.1.471.
- **75.** LeBlanc TW, Roeland EJ and El-Jawahri A. Early Palliative Care for Patients with Hematologic Malignancies: Is It Really so Difficult to Achieve? *Curr Hematol Malig Rep* 2017; 12: 300-308. 2017/06/24. DOI: 10.1007/s11899-017-0392-z.
- **76.** El-Jawahri A, Greer JA, Pirl WF, et al. Effects of Early Integrated Palliative Care on Caregivers of Patients with Lung and Gastrointestinal Cancer: A Randomized Clinical Trial. *Oncologist* 2017; 22: 1528-1534. 2017/09/13. DOI: 10.1634/theoncologist.2017-0227.
- **77.** Fulmer T, Escobedo M, Berman A, et al. Physicians' Views on Advance Care Planning and End-of-Life Care Conversations. *J Am Geriatr Soc* 2018; 66: 1201-1205. 2018/05/26. DOI: 10.1111/jgs.15374.
- **78.** Anselm AH, Palda V, Guest CB, et al. Barriers to communication regarding end-of-life care: perspectives of care providers. *J Crit Care* 2005; 20: 214-223. 2005/10/29. DOI: 10.1016/j. jcrc.2005.05.012.
- **79.** Prod'homme C, Jacquemin D, Touzet L, et al. Barriers to end-of-life discussions among hematologists: A qualitative study. *Palliat Med* 2018; 32: 1021-1029. 2018/05/15. DOI: 10.1177/0269216318759862.
- **80.** Paladino J, Brannen E, Benotti E, et al. Implementing Serious Illness Communication Processes in Primary Care: A Qualitative Study. *Am J Hosp Palliat Care* 2021; 38: 459-466. 2020/08/15. DOI: 10.1177/1049909120951095.
- Slort W, Blankenstein AH, Deliens L, et al. Facilitators and barriers for GP-patient communication in palliative care: a qualitative study among GPs, patients, and end-oflife consultants. Br J Gen Pract 2011; 61: 167-172. 2011/03/29. DOI: 10.3399/bjgp11X567081.
- **82.** Flierman I, Nugteren IC, van Seben R, et al. How do hospital-based nurses and physicians identify the palliative phase in their patients and what difficulties exist? A qualitative interview study. *BMC Palliat Care* 2019; 18: 54. 2019/07/11. DOI: 10.1186/s12904-019-0439-0.
- **83.** Bruera E, Kuehn N, Miller MJ, et al. The Edmonton Symptom Assessment System (ESAS): a simple method for the assessment of palliative care patients. *J Palliat Care* 1991; 7: 6-9. 1991/01/01.
- **84.** Brooks JV, Poague C, Formagini T, et al. The Role of a Symptom Assessment Tool in Shaping Patient-Physician Communication in Palliative Care. *J Pain Symptom Manage* 2020; 59: 30-38. 2019/09/09. DOI: 10.1016/j.jpainsymman.2019.08.024.
- **85.** Stromgren AS, Sjogren P, Goldschmidt D, et al. Symptom priority and course of symptomatology in specialized palliative care. *J Pain Symptom Manage* 2006; 31: 199-206. 2006/03/28. DOI: 10.1016/j.jpainsymman.2005.07.007.
- 86. Field D. Formal teaching about death and dying in UK medical schools. 1984: 429-434.
- 87. Field D. Education for palliative care: formal education about death, dying and bereavement in UK medical schools in 1983 and 1994. *Medical Education* 1995; 29: 414-419.

- **88.** Walker S, Gibbins J, Paes P, et al. Palliative care education for medical students: Differences in course evolution, organisation, evaluation and funding: A survey of all UK medical schools. *Palliat Med* 2017; 31: 575-581. 2017/04/26. DOI: 10.1177/0269216316671279.
- **89.** Field D and Wee B. Preparation for palliative care: teaching about death, dying and bereavement in UK medical schools 2000-2001. *Medical Education* 2002; 36: 561-567.
- **90.** Eychmüller S, Forster M, Gudat H, et al. Undergraduate palliative care teaching in Swiss medical faculties: a nationwide survey and improved learning objectives. *BMC medical education* 2015; 15: 213. DOI: 10.1186/s12909-015-0485-0.
- **91.** Weber M, Schmiedel S, Nauck F, et al. Knowledge and attitude of final year medical students in Germany towards palliative care an interinstitutional questionnaire-based study. *BMC palliative care* 2011; 10: 19. DOI: 10.1186/1472-684X-10-19.
- **92.** Ilse B, Alt-Epping B, Kiesewetter I, et al. Undergraduate education in palliative medicine in Germany: a longitudinal perspective on curricular and infrastructural development. *BMC Med Educ* 2015; 15: 151. 2015/09/19. DOI: 10.1186/s12909-015-0439-6.
- **93.** Lorig K. Self-Management of Chronic Illness: A Model for the Future. *Generations: Journal of the American Society on Aging* 1993; 17: 11-14.
- **94.** Abma T, Banks S, Cook T, et al. *Participatory Research for Health and Social Well-Being.* Springer Nature Switzerland AG, 2019.
- 95. McDarby M, Mroz E, Carpenter BD, et al. A Research Agenda for the Question Prompt List in Outpatient Palliative Care. J Palliat Med 2021; 24: 1596-1597. 2021/11/03. DOI: 10.1089/ jpm.2021.0437.
- **96.** Kripp M, Willer A, Schmidt C, et al. Patients with malignant hematological disorders treated on a palliative care unit: prognostic impact of clinical factors. *Ann Hematol* 2014; 93: 317-325. 2013/08/09. DOI: 10.1007/s00277-013-1861-7.
- **97.** Kruhlak M, Kirkland SW, Clua MG, et al. An Assessment of the Management of Patients with Advanced End-Stage Illness in the Emergency Department: An Observational Cohort Study. *J Palliat Med* 2021; 24: 1840-1848. 2021/07/14. DOI: 10.1089/jpm.2021.0004.
- 98. Ouchi K, Block SD, Schonberg MA, et al. Feasibility Testing of an Emergency Department Screening Tool To Identify Older Adults Appropriate for Palliative Care Consultation. J Palliat Med 2017; 20: 69-73. 2016/09/16. DOI: 10.1089/jpm.2016.0213.
- 99. Blomaard LC, de Groot B, Lucke JA, et al. Implementation of the acutely presenting older patient (APOP) screening program in routine emergency department care : A before-after study. Z Gerontol Geriatr 2021; 54: 113-121. 2021/01/21. DOI: 10.1007/s00391-020-01837-9.
- **100.** Pieters J, Verstegen DML, Dolmans D, et al. Design and evaluation of a learning assignment in the undergraduate medical curricula on the four dimensions of care: a mixed method study. *BMC Med Educ* 2021; 21: 309. 2021/06/02. DOI: 10.1186/s12909-021-02681-0.
- **101.** Richards CT, Gisondi MA, Chang CH, et al. Palliative care symptom assessment for patients with cancer in the emergency department: validation of the Screen for Palliative and Endof-life care needs in the Emergency Department instrument. *J Palliat Med* 2011; 14: 757-764. 2011/05/10. DOI: 10.1089/jpm.2010.0456.
- 102. George N, Barrett N, McPeake L, et al. Content Validation of a Novel Screening Tool to Identify Emergency Department Patients With Significant Palliative Care Needs. Acad Emerg Med 2015; 22: 823-837. 2015/07/15. DOI: 10.1111/acem.12710.
- 103. Reuter Q, Marshall A, Zaidi H, et al. Emergency Department-Based Palliative Interventions: A Novel Approach to Palliative Care in the Emergency Department. J Palliat Med 2019 2019/02/06. DOI: 10.1089/jpm.2018.0341.
- 104. Bruera E and Hui D. Conceptual models for integrating palliative care at cancer centers. J Palliat Med 2012; 15: 1261-1269. 2012/08/29. DOI: 10.1089/jpm.2012.0147.

Ŭ